Author’s response to reviews

Title: Case report of long-term follow-up after denosumab treatment for osteoporosis: rebound associated with hypercalcaemia, parathyroid hyperplasia, severe bone mineral density loss and multiple fractures

Authors:

Yves Maugars (yves.maugars@chu-nantes.fr)
Pascale Guillot (pascale.guillot@chu-nantes.fr)
Joëlle Glémarec (joelle.glemarec@chu-nantes.fr)
Jean-Marie Berthelot (jean-marie.berthelot@chu-nantes.fr)
Benoit Le Goff (benoit.legoff@chu-nantes.fr)
Christelle Darrieutort-Laffite (christelle.darrieutort@chu-nantes.fr)

Version: 2 Date: 27 Nov 2019

Author’s response to reviews:

I have highlighted the modifications in the text.
Responses to reviewer 3

Comments 4 and 9:

We have added the details of the type of fractures (type and grade), and the MRI findings at the level of the other vertebrae than T3. The importance of the pain has been emphasized.
The time of the beginning of risedronate has been added (the one of zoledronate is in the text).

Comment 7

The opportunity of a preventive bisphosphonate therapy has been added in the abstract.

I did not really well understand the comment 5.

Responses to reviewer 4

Comment9
Updated publications added.